Author’s response to reviews

Title: FAM168A participates in the development of chronic myeloid leukemia via BCR-ABL1/AKT1/NFκB Pathway

Authors:

Xiaorong Liu (liuxiaorongsmile@yahoo.com)
Huirong Mai (maihuirong@163.com)
Hanfang Jiang (hanfang923@163.com)
Zhihao Xing (xingzhihao@hotmail.com)
Dong Peng (donnapd@163.com)
Yuan Kong (kongyuan1019@126.com)
Chunqing Zhu (1052796138@qq.com)
Yunsheng Chen (chenyunshenglw@163.com)

Version: 3 Date: 27 Jun 2019

Author’s response to reviews:

Responses letter

Note: the numbers of the lines refer to these in the revised manuscript.

Technical Comments:

1. We notice that there is still identifying information on the patients in your manuscript, in particular in the 'Materials' subsection of the Methods - "Three patients, 2 males (6.5 and 8 years old) and 1 female (8.5 years old)". Please provide this information as age ranges.

Re: in the line 120 “6 – 10” instead of “6.5 and 8” and “8.5”. Thank you.
2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Re: a single, final, clean version was uploaded. Thank you.